CENWP-OP-B                                                                                                                        17 August 2003

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 33 of 2003, which covers the period 10 through 16 August 2003. 

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 131.6 to 172.6 kcfs.  Daily project tailwater elevation averaged 10.5’ to 13.4’ msl.  Daily project forebay elevation averaged 75.0’ to 75.9’ msl.  River temperature averaged 69 to 70° F.  Daily turbidity measured 5.5’ to 7.0’ on the secchi disk.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascade Island and Bradford Island fishways respectively, are included in the spill pattern.  Spill averaged 84.7 to 96.3 kcfs.  

1.3 Unit Operation

1.3.1 Unit 1 remains out of service for turbine rehab.

1.3.2 Unit 2 remains out of service due to crane rail replacement.  
1.3.3 Unit 3 remains out of service due to crane rail replacement.  

1.3.4 Unit 4 remains out of service due to crane rail replacement. 
1.3.5 Unit 5 remains out of service due to crane rail replacement. 

1.3.6 Unit 6 remains out of service due to crane rail replacement. 

1.3.7 Unit 7 remains out of service for its five year overhaul.

1.3.8 Unit 8 remains out of service due to crane rail replacement. 

1.3.9 Unit 14 remains out of service for its five year overhaul.

1.3.10 Unit 18 was out of service from 0754 until 0918 on 13 August for STSs repairs.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  F2 was at reduced load from 1341 until 1609 on 12 August to facilitate trashrack raking activities.

1.5 Gatewell drawdowns are measured on Wednesdays.

2.  ACTIVITIES. 
2.1 AWS Closures.  No AWS valves were closed this reporting week.

2.2 STS/VBS Inspections.  The following STS/VBS inspections were conducted this reporting week.

	Unit
	Date
	Run Hours
	Comments

	11
	8/11
	869
	

	12
	8/11
	579
	

	13
	8/11
	497
	

	14
	8/11
	314
	Pulled for unit overhaul

	15
	8/11
	544
	

	16
	8/11
	501
	

	17
	8/11
	456
	

	18
	8/11
	971
	18” X 2” rip in A slot screen.  Repaired on 13 August.


Table 1.  STS Inspection Dates.

2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage activities occurred this week.  

3.  RESEARCH.

3.1 CRITFC.  Sampling has concluded until water levels drop below 70º F.

3.2 NMFS Lamprey.  Trapping, tagging and tracking continue.  See misc. 6.3
3.3 ODFW Steelhead.  Trapping continues.  See misc. 6.3
3.4 PSMFC at the SMF.  GBT exams continue.  

3.5 University of Idaho.  Adult tracking tagging and tracking continue.  See misc. 6.3
3.6 WDFW Fish Counting.  16-hr visual counts at Washington Shore and Bradford Island.

3.7 WDFW Tule counter.  Visual counts start at Washington Shore and Bradford Island started 15 August.

4.  FISHWAYS 
4.1 Project biologists inspected 10 through 16 August this reporting week.  NMFS inspected on 15 August.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	PH1 PCC
	Velocity <1.5fps
	Meter in slowest part of channel
	No response at this time.
	

	All week
	PH2 PCC
	No velocity measure.
	Cable not installed.
	See misc. 6.1
	

	8/10
	Weir 37
	Reading < 1.3’
	Unknown
	No response at this time.
	

	All week
	PH1 ITS
	Chain gate 7A and ITS end gate closed.
	Crane rail work
	8B opened.  See const. 5.2.1
	

	8/15-16
	Weir 37, Weir 38, UMT weir, A and B branch weirs
	Reading > 1.0’ +/- .1’
	Not yet changed out of shad passage mode.
	
	#’s 50563, 50564, 50565

	All week
	Bonneville fishways
	Calibration not checked by fisheries.
	
	
	


4.3 Fish Ladder Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperatures are in Celsius this week.  http://students.washington.edu/kyle/temptable.jpg   (295.37K = 22.22C= 72F)

	Date
	WA Shore
	Bradford Is.
	AFF exit

	8/10
	22.15
	22.15
	22.15

	8/11
	22.15
	22.05
	22.15

	8/12
	22.15
	22.15
	22.15

	8/13
	22.05
	22.05
	22.05

	8/14
	22.05
	22.05
	22.05

	8/15
	21.95
	21.95
	22.05

	8/16
	21.85
	21.85
	21.85


4.4 Sensor Calibration.  Calibration was not checked this reporting week.

4.5  Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. CONSTRUCTION

5.1 PH 2 Corner Collector

5.1.1 In a 10 July teletype the Project was given a hard constraint maximum tailwater of 16’ until 01 September.  This is for the B2CC work at PH2.  In a 15 July teletype, the maximum tailwater constraint overrules spill to the 120% gas cap if spill raises tailwater over 16’.

5.2 PH 1 Crane Rail Replacement- started on 01 August and will continue until 16 December.
5.2.1 As a safety precaution, chain gate 7A has been closed (8A to compensate) and the end gate will be closed during the hours of 0530 until 1900 for the contractors working over the ITS.  The end gate will be opened each evening to flush fish.  

5.2.2 On 12 August, a teletype was issued indicating unit priority for PH1 during crane rail replacement.  Until 16 September, Unit 9 is to run before Unit 10 and both units are to have screens pulled.  After 16 September, Unit 10 is to run before Unit 9 and screens must be installed.  If either unit operates, the ITS will be opened and allowed to run for not less than four hours after the units have been shutdown.  This is to flush any fish that may have been attracted into PH1 forebay.

5.2.3 PH1 ITS was closed from; 0525 until 1655 on 11 August, 0530 until 1648 on 12 August, 0530 until 1620 on 13 August, 0612 until 1615 on 14 August, 0521 until 1700 on 15 August, 0539 until 1850 on 16 August.

6. MISCELLANEOUS

6.1 A new velocity meter was installed for PH2 PCC during the mid season dive on 21 July.  We are now waiting for the electricians to finish installation.

6.2 As of 28 June, the Project will have a daytime spill of 75 kcfs.

6.3 Water temperatures remain near 72ºF this reporting week.  Researchers trapped early in the week, closing the AFF down before noon.  They also trapped on 13 August.  Temperatures appear to be dropping.  Researchers will be in four days a week if temperatures are between 70ºF and 72ºF.

6.4 FV4-3 and FV4-4 were put in manual at 1315 on 16 August.  The computer is not programmed for the low tailwater limits.

JAMES R. MAHAR, P.E.

Operations Manager
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